Author’s response to reviews

Title: Factors Associated with Quality of Life of Postmenopausal Women Living in Iran

Authors:

Soheila Nazarpour (sn_9547@yahoo.com)
Masoumeh Simbar (msimbar@gmail.com)
Fahimeh Ramezani Tehrani (fah.tehrani@gmail.com)
Hamid Alavi Majd (alavimajd@gmail.com)

Version: 2 Date: 04 Feb 2020

Author’s response to reviews:

Dear Assistant Editor

We sincerely appreciate your valuable comments on our manuscript entitled: “Factors Associated with Quality of Life of Postmenopausal Women Living in Iran”. We have provided a revised version based on these comments, along with a response (R) to the comments and hope that the response meet your expectations.

Assistant Editor Comments:

1. Overlap
   --Thank you for your response to our previous revisions request. However, we must stress that we cannot condone the use of previously published text even if this is work you have previously published.
   For methods and studies you have previously done, please summarise and reference the original study instead.

   R- The method section was summarized and cited to the previous published study.

   Please also revise the 'Strengths and Limitations' subsection of your 'Discussion' to reduce the possible overlap with your previous study.

   R- The 'Strengths and Limitations' subsection was revised

Best regards.